Author's response to reviews

Title: Knowledge and attitudes of primary health care physicians and nurses with regard to population screening for colorectal cancer in Balearic Islands and Barcelona.

Authors:

Maria Ramos (mramos@dgsanita.caib.es)
Magdalena Esteva (mesteva@ibsalut.caib.es)
Jesus Almeda (jameda@ambitcp.catsalut.net)
Elena Cabeza (ecabeza@dgsanita.caib.es)
Diana Puente (dpuente@idiapigol.org)
Rosa Saladich (rsaladich@gmail.com)
Albert Boada (aboadav.bcn.ics@gencat.cat)
Maria Llagostera (mllagostera.cp.ics@gencat.cat)

Version: 7 Date: 10 September 2010

Author's response to reviews: see over
Srs. Editors BMC Cancer:

I submit you the manuscript “Knowledge and attitudes of primary health care physicians and nurses with regard to population screening for colorectal cancer” reviewed according formatting changes required. The following changes have been made:

- The list of authors has been written following the submission system.
- We tried to include the tables within the text, although we have had some problems moving the legend below the tables, as you can see. We hope you can correct it.
- We changed the way references are cited in the text using square brackets.
- We have removed highlighting and tracking from the manuscript.
- We have reviewed the references following the BMC Cancer style.
- Finally, we reviewed the typography.

Maria Ramos